
Mauna Loa Solar Observatory Observer's Log

Tue Feb 22 17:56:49 GMT 2011

Year: 11 Doy: 053

Observer: koon

WEATHER COMMENT: Tue Feb 22 17:56:50 GMT 2011

Thick altocumlus and cirrostratus overcast, no wind, temp=45F, no obs until

clouds thin out.

____end___

NOTE BY DARRYL: Tue Feb 22 21:14:49 GMT 2011

HVO is now on the new network, that's it, everyone has been moved onto the new network, I'll turn off the old router and let people know that we want to cancel the old network connection with HawTel and UH. Yeehaw! We're still using the old switch as a temporary 10/100 Mbps media converter for GONG via their old fiber optic cable, that will be changed in the near future when GONG sends us either patch cables for the new shelter media converters or a new shelter network switch that is Gigabit ready and outputs fiber optic cables via SFP connectors.

___end___

WEATHER COMMENT: Tue Feb 22 21:16:31 GMT 2011

Still very cloudy.

____end___

Tue Feb 22 23:32:59 GMT 2011

MkIV